Author's response to reviews

Title: Leukocytosis and high hematocrit levels yield misinterpretation in abdominal emergency of hereditary angioedema (HAE)

Authors:

Isao Ohsawa (iohsawa@juntendo.ac.jp)
Seiji Nagamachi (snagama@juntendo.ac.jp)
Hiyori Suzuki (hiyori@juntendo.ac.jp)
Keisuke Honda (dhonda@juntendo.ac.jp)
Nobuyuki Sato (nobsato@juntendo.ac.jp)
Hiroyuki Ohi (ohi@h8.dion.ne.jp)
Satoshi Horikoshi (satoshi@juntendo.ac.jp)
Yasuhiko Tomino (yasu@juntendo.ac.jp)

Version: 4 Date: 25 October 2012

Author's response to reviews: see over
Dear Editor:

“Leukocytosis and high hematocrit levels yield misinterpretation in abdominal emergency of hereditary angioedema (HAE)” by Isao Ohsawa, et al; ms. no 2697156368287584.
We are most grateful to you and have addressed a point-by-point description of the changes.

Since we had received informed consent from each patient prior to this study, we added the sentence of “A written informed consent was given by all participating patients.” in the end of the first section of “Patients and Methods”,

We hope that the revised version of our paper is now suitable for publication in the “BMC Gastroenterology” and we look forward to hearing from you at your earliest convenience.

Sincerely yours,
Isao Ohsawa M.D.